Author's response to reviews

Title: Patients' functioning as predictor of nursing workload in acute hospital units providing rehabilitation care: a multi-centre cohort study

Authors:

    Martin Mueller (martin.mueller@med.uni-muenchen.de)
    Stefanie Lohmann (stefanie.lohmann@med.uni-muenchen.de)
    Ralf Strobl (ralf.strobl@med.uni-muenchen.de)
    Christine Boldt (christine.boldt@paranet.ch)
    Eva Grill (eva.grill@med.uni-muenchen.de)

Version: 6 Date: 27 October 2010

Author's response to reviews:

We have relaced a ',' by a '.' in the email-address of co-author Stefanie Lohmann (p. 1)

Since there is a new directive of our organization, we have to replace
'Ludwig-Maximilians-University' by 'Ludwig-Maximilians-Universität' in the
affiliations (p. 1)